Author's response to reviews

Title: Knowledge and Utilization of Information Communication Technology (ICT) among Health Science Students at the University of Gondar, North Western Ethiopia.

Authors:

Solomon Assefa (solomonazezo@gmail.com)
Yigzaw Kebede (gkyigzaw@yahoo.com)
Desalegn T Zegeye (zdesalegn@gmail.com)

Version: 3 Date: 11 February 2013

Author's response to reviews: see over
Author’s response to reviews

Title: Knowledge and Utilization of Information Communication Technology (ICT) among Health Science Students at the University of Gondar, North Western Ethiopia.

Authors:

Solomon Assefa: solomonazezo@gmail.com

Yigzaw Kebede: gkyigzaw@yahoo.com

Desalegn T Zegeye: zdesalegn@gmail.com

Version: 2

Date: 11/02/2013

Author's response to reviews: see over

Dear Adrian Aldcroft

We wanted to thank you and the reviewers for your time in reviewing our manuscript and the thoughtful suggestions. We feel our revised manuscript is improved because of the reviewer’s suggestions and hope you find these changes acceptable.

Thank you.